DRAFT:
 Notes of the RICH1 Phone meeting 29/04//2005
People Present:

Bristol: Richard

CERN:  Fabio, Nick, Werner 
IC: Bill, Dave, Geoffrey, Trevor 

Oxford: excused
RAL: Mike

Apologies for absence:

CERN: Carmelo
Discussed Agenda Topics:
Note: A status report for most of the agenda topics has already been given at the RICH group meeting on the 27/04/05, see the CDS Agenda.

1. Magnetic Shielding: 
1.1 Status:

The magnetic shield components are bolted together and are placed at the top of the LHCb access shaft. They are ready to be moved into the pit. All rails/pieces are there, except for the bracket .The support concrete beam to bolt into the wall arrived late. Work schedule could be delayed by one week. 
1.2 Magnetic Field Measurements:
The magnetic field measurements are probably delayed by one week: start ~20th May. The B-machine will be kept a bit longer than planned and then will be given to ALICE. 

2. Gas Enclosure:
The NAB company, located north of London, has been selected, because of the lower costs. Both Tony and Rohan have received weld samples and visited the company previously (quite happy). Paper work is in preparation and it is still possible to add notes for updating. The order is to be sent out within three weeks [next week (first week of May)]. The NAB time scale from the order to the completion of the gas enclosure is 10-12 weeks.

There is still some minor remaining work to do on the gas enclosure e.g. quartz window, etc.
At the meeting at IC with Rohan (18/4/05) the basic design/concept for the gas enclosure support/handling has been discussed and is moving ahead.
3. VELO Seal
Mike could find no company that could produce aluminum bellows, therefore stainless steel bellows will be used. Drawings are on EDMS.  No response from the VELO group regarding the drawings. Dave will send a mail reminder to Massi regarding the drawings (action Dave).
Regarding the VELO dimension change: the 6mm movement is on the UPSTREAM flange (i.e. the other side of the IP), so it has no effect on RICH1.  Due to shrinking of the welds, the VELO vessel is shorter than on the drawings. The position of the exit foil flange is moved 1.5 mm upstream and the upstream flange is moved 6 mm downstream (closer to z).

Plan to have the VELO seal PRR together with the Exit Window.


4.  Exit Window
Drawings to be put in EDMS by the beginning of next week (action Mike). There is a new design: one piece instead of four, the composite skins are made from one skin each. This is an easier and stabler design. A company in India can produce the aluminum diaphragms in 4-6 weeks. RAL will get some samples in ~2 weeks for testing (thickness). Once tested the diaphragms will be ordered from India with the appropriate thickness.

The beam-pipe meeting is on Tuesday 17 May at CERN (9:00 in room B30/2-005).
5. Beryllium Mirrors
The status report was given at the RICH group meeting (27/04/05).
Prototype: The prototype was sent back to Moscow (Kompozit) to correct the radius of curvature by annealing. Now it is in St. Petersburg for vitrification. Then it should be grinded and polished. An estimate for its delivery could be known sometime next week.  
2nd Mirror: order was sent for blank
Be-Production: ISTC preliminary documents ready. According to the schedule the last mirror could be delivered at CERN by June 06.
Drawings: No feedback news from Vladimir K. regarding the UIR drawing. 
Backup Plan:

The backup plans were presented at the RICH group meeting (27/04/05). There are two plans. A temporary backup using composite mirrors (Tito’s talk). The mirrors could be ready in 10 months for a reasonable price, but there would be 4 mirrors per quadrant. The permanent backup solution is based on using CMA mirrors (Fabio’s talk), with one mirror per quadrant. These mirrors are relatively expensive and would need to be tested again. 
6. Flat Mirrors
RICH2 has received good quality mirrors from the Czech company Compass. As soon as the RICH2 flat mirrors are completed/delivered, an order for the RICH1 mirrors will be placed at the same company. Mirror thickness ~7mm.

The mirror size has still to be resolved (action Dave+Richard).
Natural frequency of the mirrors ( 6-7mm): check with RICH2  (action Fabio). 

7. Aerogel
No news, still need to resolve conflicts.
9. Installation and Handling

Work is underway.  To be discussed in detail at the project overview meeting (3rd of May)
Geof Barber will be working fulltime for a period of 2 months on the RICH1 mirror mounting/installation design together with Richard.
10. RICH1 Project Overview Meeting
To be held on the 3rd of May at CERN in the afternoon at 15:00 (CERN time).  The room 40/4-C01 with VRVS and phone has been reserved. 
11.  Agenda/Notes on CDS
The RICH1 phone-in meeting agenda and notes are on CDS.
12. Next meeting
Friday  13th May at 10:00(GMT) 11:00(CET). 

